Author’s response to reviews

Title: Changes in Diet and Physical Activity Resulting from the Shape Up Somerville Community Intervention

Authors:

Sara C Folta (sara.folta@tufts.edu)
Julia F Kuder (julia.kuder@gmail.com)
Jeanne P Goldberg (jeanne.goldberg@tufts.edu)
Raymond R Hyatt (raymond.hyatt@tufts.edu)
Aviva Must (aviva.must@tufts.edu)
Elena N Naumova (elena.naumova@tufts.edu)
Miriam E Nelson (miriam.nelson@tufts.edu)
Christina D Economos (christina.economos@tufts.edu)

Version: 4 Date: 6 September 2013

Author’s response to reviews: see over
6 September 2013

Dear BMC Pediatrics Editorial Board:

Accompanying this letter please find a revised version of our article, “Changes in Diet and Physical Activity Resulting from the Shape Up Somerville Community Intervention”, manuscript #7610741489151498 which we are submitting as a Research Article. We have made the minor revisions requested by Referee #1, as follows:

- **The authors should add “years” after the age of children. For example, children were aged 6-11 years.** We have added “years” after the age in all instances within the manuscript.
- **On page 9 at the top, the authors might wish to include a single sentence indicating if activities were for the most part implemented as intended as discovered by the process evaluation.** We have added this sentence to the manuscript.

We very much appreciate the review and consideration for publication in BMC Pediatrics.

Sincerely,


Sara C. Folta, PhD

Corresponding Author

Julia F. Kuder, MA
Jeanne P. Goldberg, PhD, RD
Raymond R. Hyatt, PhD
Aviva Must, PhD
Elena N. Naumova, PhD
Miriam E. Nelson, PhD
Christina D. Economos, PhD